Author's response to reviews

Title: Endoscopic Saphenous harvesting with an Open CO2 System (ESOS) Trial for coronary artery bypass grafting surgery: study protocol for a randomized controlled trial.

Authors:

Antonio Campanella MD (antoniocampanella@libero.it)
Laura Bergamasco PhD (laura.bergamasco@unito.it)
Luigia Macri’ MD (lmacri@molinette.piemonte.it)
Sofia Asioli MD (sofia.asioli@unito.it)
Roger Devotini MD (rogerdevotini@hotmail.com)
Silvana Barbaro Dr (sbarbaro@molinette.piemonte.it)
Serenella Scipioni Dr (sscipioni@molinette.piemonte.it)
Pietro Rispoli MD (pietro.rispoli@unito.it)
Mauro Rinaldi MD (mauro.rinaldi@unito.it)

Version: 7 Date: 14 November 2011

Author's response to reviews:

Dear Editors,
Thank you for your email and for your suggestions.
We have checked our manuscript carefully for any typing or content errors. These will be included in the revised version and these will be highlighted in yellow.
Please contact us if you have any problems or questions.
Kind Regards